Author's response to reviews

Title: Broad autism phenotype status of parents of children with autism in China and its correlation with severity of social impairment in probands

Authors:

Lijuan Shi (zijuanwd@163.com)
Jianjun Ou (oujianjun08@163.com)
Jingbo Gong (yefenhua@126.com)
Suhong Wang (yarmine@foxmail.com)
Yuanyue Zhou (taoyoyo1977@yeah.net)
Furong Zhu (zhufurong1985@126.com)
Xudong Liu (liux@queensu.ca)
Jingping Zhao (zhaojingpinghunan@163.com)
Xuerong Luo (luoxr@vip.sina.com)

Version: 2
Date: 22 July 2014

Author's response to reviews: see over
Dear Editor,

Thank you very much for your attention and your advices on our paper “Broad autism phenotype status of parents of children with autism in China and its correlation with severity of social impairment in probands”. We have changed two references in the Discussion part, modified Table 4 and pages and lines were numbered in this revised manuscript.

We sincerely hope this manuscript will be in the peer review process as soon as possible.

Thank you very much for your help and looking forward to hearing from your soon.

Best regards,

Sincerely yours,

Xue-Rong Luo